Author's response to reviews

Title: Infiltrating ductal carcinoma breast with central necrosis closely mimicking ductal carcinoma in situ (comedo type): a case report

Authors:

Shahid Pervez (shahid.pervez@aku.edu)
Hassan Khan (drhasankhan@gmail.com)

Version: 6 Date: 3 September 2007

Author's response to reviews:

Manuscript is now reformatted as per suggestion (your mail dated August 30, 2007 2:49 PM).
However we still suggest that title may have 'A case series' instead of 'A case report' as this manuscript describes four cases.

Regards,

Dr Shahid Pervez